% Example of use of the li-article style
% B. Sagot, 2008-2013
% E. Laporte, 2014
\documentclass[fleqn,english]{li-article}


% The 'mathptmx' package will change the default roman font family to Times,
% and the virtual 'mathptmx' fonts will be used for math.
% In case of problems, use \usepackage{times}
% in place of \usepackage{mathptmx}
\usepackage{mathptmx}

\usepackage{graphicx}

\journal{Lingvistic{\ae} Investigationes. Volume nn -- n° x/2014}{00}{00}

% The title of your article
\title{Full title of the paper}

\author[F. LastName1, F. LastName2, F. LastName3]{FirstName1 LastName1, FirstName2 LastName2, FirstName3 LastName3}

\institution{University of Here - University of There}

\address{FirstName1 LastName1, FirstName2 LastName2\\
Lab1, University of Here\\
5 boulevard Descartes - Champs-sur-Marne\\
77454 Marne-la-Vall\'ee Cedex 2, France\\
\\
FirstName3 LastName3\\
Lab2, University of There\\
30 rue du Ch. des rentiers\\
75013 Paris, France
}

\abstract{Abstract in English.}

\begin{document}

\maketitle

\section*{Introduction}

The {\tt \textbackslash journal} command must be completed with the appropriate volume and number
information.  The last two arguments of this command are the first and last page numbers, which
allows the authors to paginate correctly.

The short title is used for the headers of even pages, the short version of the authors list is used
for the headers of the odd pages. These short versions are defined by the {\tt \textbackslash {\em
command}[short version]\{full version\}} command, where {\tt\em commande} is respectively
{\tt title} or {\tt author}. If the short version of one of these pieces of information is not defined,
the full version is used instead. However, the text used for headers must not be too long (less than
80mm); if it is the case, the compilation of the document fails, with a reasonable error
message \cite{complex99,Keenan-1976}.

The first paragraph of a (sub)section is not indented \cite{Molinier00}, the following
are \cite{baptista-ranchhod-1998}.

Lists look as follows:
\begin{itemize}
\item item 1
\item item 2
\end{itemize}

\section{Section}

Introduction of the section, with a little linguistic example \cite{Chaurand00,de-gioia-1994}.

\enumsentence{\textit{Examples are obligatorily in italics} }

\subsection{Sub-section}

Introduction of the sub-section \cite{gross-1981}.

\eenumsentence{
\item[a.] \textit{Ceci est un exemple}\\ ``This is an example''
\item[b.] *\textit{Ceci un exemple}\\ *\textit{This an example}
}

\subsubsection{Sub-sub-section}

Content of the sub-sub-section.

%\nocite{*}

%%===================================================================

\bibliography{li-example}

%%===================================================================

\makesummary
\makeaddress

%%===================================================================
\end{document}
